Author’s response to reviews

Title: Family Psychosocial Assessment in the Clinical Setting

Authors:

Arwa Nasir (anasir@unmc.edu)
Andrea Zimmer (andrea.zimmer@boystown.org)
David Taylor (david.taylor@unmc.edu)
Jonathan Santo (jsanto@unomaha.edu)

Version: 2 Date: 11 Jul 2018

Author’s response to reviews:

Dear Editor,

Thank you for your review of our manuscript: Psychosocial Assessment of the Family in the Clinical Setting.

We revised the manuscript according to the reviewers’ recommendations. Please find below a point by point response. The changes are highlighted in the text.

1. Description of the properties of the PSC-16 moved to the methods section.

2. We specified the properties of the validation measures.

3. Justification of the sample size is provided including reference.

4. In page 7, references are provided to support the effect of the psychosocial factors included in the FHQ.

5. All figures are included in the full submission.
6. Statistical analysis section is added to the methods section.

Thank you again for your review. Please let me know if you have any further questions or comments.

Best regards,